Author’s response to reviews

Title: A literature review and case report of severe and refractory post-colectomy enteritis

Authors:

Yingyun Yang (yingyunyang@126.com)
Yuan Liu (ly880817@126.com)
Weiyang Zheng (oven186@126.com)
Weixun Zhou (zweixun@163.com)
Bin Wu (wubin0279@hotmail.com)
Xiyu Sun (sunpumch@126.com)
Wei Chen (chenw@pumch.cn)
Tao Guo (guoqiong990@126.com)
Xiaoqing Li (lixiaoqing20060417@126.com)
Hong Yang (hongy72@163.com)
Jiaming Qian (qianjiaming1957@126.com)
Yue Li (yuelee76@gmail.com)

Version: 2 Date: 28 Mar 2019

Author’s response to reviews:

Dear Editors,

Thank you so much for considering our paper and providing the essential revision suggestions for us to improve our manuscript! And great thanks to the reviewers for their efforts! We have made the revisions suggested, including asking the language service to reedit our manuscript. Hope it is improved.
Thank you again for your help!

Yue Li